Author’s response to reviews

Title: A case report of Pachydermoperiostosis (Touraine-Solente-Gole syndrome) from Nepal

Authors:
Amir Joshi (amir.joshi33@iom.edu.np)
Gaurav Nepal (gauravnepal@iom.edu.np)
Yow Shing (yow_ks@yahoo.com.sg)
Hari Prasad Panthi (haripp@gmail.com)
Suman (sumanbaral@yahoo.com)

Version: 2 Date: 18 Dec 2018

Author’s response to reviews:

Additional revisions are needed.
Ans: We made the changes.

Give information about follow-up for at least 6 months.
Ans: We added the information about follow-up.

Discussion – add a paragraph at the beginning of the Discussion that summarizes the case and describes what is unique in this case compared to what is available in the literature.
Ans: We made the changes as suggested by editor.